Author's response to reviews

Title: Communicating effectiveness of intervention for chronic diseases: what single format can replace comprehensive information?

Authors:

Henrik Stovring (hstovring@health.sdu.dk)
Dorte Gyrd-Hansen (dgh@dsi.dk)
Ivar S Kristiansen (ivarск@c2i.net)
Jørgen Nexøe (jnexoe@health.sdu.dk)
Jesper B Nielsen (jbnelsen@health.sdu.dk)

Version: 9 Date: 17 June 2008

Author's response to reviews:

Dear Editor,

Following your instructions I have modified the title page (author names, addresses and emails), and I have reformatted all reference numbers (from superscript to square brackets).

Regarding author names, I have generally been unable to provide names with Danish vowels, ie. my surname is really Støvring, and a co-author's name is really Jørgen Nexøe. Is there any way to keep these special vowels? If not, the current spelling is the correct transcoding (ø is replaced by o).

Regards,

Henrik